Author's response to reviews

Title: High quality copy number and genotype data from FFPE samples using Molecular Inversion Probe (MIP) microarrays

Authors:

Yuker Wang (yuker_wang@affymetrix.com)
Victoria E.H. Carlton (victoria_carlton@affymetrix.com)
George Karlin-Neumann (george_karlin-neumann@affymetrix.com)
Ronald Sapolsky (ron_sapolsky@affymetrix.com)
Li Zhang (lzhangli@mdanderson.org)
Martin Moorhead (martin_moorhead@affymetrix.com)
Zhigang C Wang (zhigang_wang@dfci.harvard.edu)
Andrea L Richardson (andrea_richardson@dfci.harvard.edu)
Robert Warren (warrenr@surgery.ucsf.edu)
Axel Walther (axel.walther@cancer.org.uk)
Melissa Bondy (mbondy@mdanderson.org)
Aysegul Sahin (asahin@mdanderson.org)
Ralf Krahe (rkrahe@mdanderson.org)
Musaffe Tuna (mtuna@mdanderson.org)
Patricia A Thompson (pthompson@azcc.arizona.edu)
Paul T Spellman (ptspellman@lbl.gov)
Joe W Gray (jwgray@lbl.gov)
Gordon B Mills (gmills@mdanderson.org)
Malek Faham (malek_faham@affymetrix.com)

Version: 4 Date: 22 January 2009

Author's response to reviews:

Dear Andrea,

We have deposited the data in GEO and include the reference number. I've slightly changed the format (text and figures) to match that required by BMC, have added sections for: authors' contributions, competing interests, and additional files. I've fixed minor grammatical errors (ensuring noun/verb matching, etc.), fixed an error with the references and placed in "BioMed Central" format with Endnote 5.

Regards,

Victoria